Author’s response to reviews

Title: Hypothyroidism Among Military Infants Born in Countries of Varied Iodine Nutrition Status

Authors:

Marcus M. Cranston (marcus.cranston@med.navy.mil)
Margaret A.K. Ryan (margaret.ryan@med.navy.mil)
Tyler C. Smith (tyler.c.smith@med.navy.mil)
Carter J. Sevick (carter.sevick@med.navy.mil)
Stephanie K. Brodine (stephanie.brodine@med.navy.mil)

Version: 4 Date: 25 January 2010

Author’s response to reviews: see over
Sabina Alam, PhD
Senior Scientific Editor
BMC series journals

Dr. Alam:

This letter is in response to the request to submit revisions to our article “Hypothyroidism among military infants born in countries of varied iodine nutrition status”. Please find below, the point-by-point description of the changes made:

Major revisions:

1. The abstract in the submission system will be updated to match the manuscript.

Minor revisions:

1. The title was changed to sentence case.
2. The country was included in the affiliation details.
3. The format was changed for email addresses.
4. The keywords were removed from the abstract.
5. Full citation details were included for reference 2 and 13.
6. The visible vertical lines were removed from the tables.

Sincerely,

Marcus M. Cranston, MD, MPH